Author's response to reviews

Title: Hospital staff education on severe sepsis/septic shock and hospital mortality: An original hypothesis

Authors:

Maurizia Capuzzo (cpm@unife.it)
Marco Rambaldi (m.rambaldi@ausl.mo.it)
Giovanni Pinelli (g.pinelli@ausl.mo.it)
Manuela Campesato (emanuela.campesato@virgilio.it)
Antonia Pigna (a.pigna@aosp.bo.it)
Marco Zanello (Marco.Zanello@ausl.bologna.it)
Maria Barbagallo (mbarbagallo@ao.pr.it)
Massimo Girardis (massimo.girardis@unimore.it)
Elena Toschi (elena.toschi@tiscali.it)

Version: 4 Date: 24 October 2012

Author's response to reviews:

Dear Sir,

We send you the revised version of our manuscript, with information about the Ethics Committees approvals.

We thank again you and reviewers for the work done.

Maurizia Capuzzo on behalf of all the authors